Author's response to reviews

Title: The prevalence and correlates of alcohol use and alcohol use disorders: a population based study in Sri Lanka

Authors:

Helena M.S. Zavos (helena.zavos@kcl.ac.uk)
Sisira Siribaddana (nipuna@stmail.lk)
Harriet A Ball (harriet.ball@gmail.com)
Micheal T Lynskey (micheal.lynskey@kcl.ac.uk)
Athula Sumathipala (athula.sumathipala@kcl.ac.uk)
Fruhling V Rijsdijk (fruhling.rijsdijk@kcl.ac.uk)
Matthew Hotopf (matthew.hotopf@kcl.ac.uk)

Version: 2 Date: 14 November 2014

Author's response to reviews:

In line with a request from the editorial office the manuscript now contains the following:
1. Statement of completing interests
2. Authorship contribution